Author's response to reviews

Title: Effectiveness of the Chest Pain Choice decision aid in emergency department patients with low risk chest pain: study protocol for a multicenter randomized trial

Authors:

Ryan T Anderson (anderson.ryan1@mayo.edu)
Victor M Montori (montori.victor@mayo.edu)
Nilay D Shah (shah.nilay@mayo.edu)
Henry H Ting (ting.henry@mayo.edu)
Laurie Pencille (pencille.laurie@mayo.edu)
Michele Demers (famille.leblanc.demers@gmail.com)
Jeffrey A Kline (Jefkline@iupui.edu)
Deborah B Diercks (dbdiercks@ucdavis.edu)
Judd E Hollander (Judd.Hollander@uphs.upenn.edu)
Carlos Torres (Torres.Carlos@mayo.edu)
Jason T Schaffer (jitschaff@iupui.edu)
Jeph Herrin (jeph.herrin@yale.edu)
Megan E Branda (branda.megan@mayo.edu)
Annie Leblanc (Lelanc.Annie@mayo.edu)
Erik P Hess (hess.ann@mayo.edu)

Version: 2
Date: 3 February 2014

Author's response to reviews:

Thank you for the suggested revisions. This submission includes:

1) An additional title and legend section after the reference list describing the separately uploaded Appendices 1-7.

2) The names of all ethical bodies that approved the study in each of the centers involved included in the methods section of the manuscript.